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•
 

FAR DETECTOR:
 

Rik

 

Gran
•

 

99.7% livetime

 

last week
•Partial power on Sunday outage effects surface chiller

 

pad 
only, crew called in and resets things before coil gets too hot

•
 

NEAR DETECTOR:
 

Alec Habig   
•

 

99.0% live, Smooth running for first time in a while
•Temperature has increased as the rock warms back up, but 
still plenty of headroom with the new LCW system
•Coil placed under ACNET control

•

 

Preliminary look at NuMI

 

spectrum validates model of the 
effect of reduced cooling on the neutrino spectrum
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•

 

BEAM SYSTEM STATUS:

•

 

Smooth 11-batch running except 
for NuMI

 

accesses on T+W (6.5 
extra hours), TeV

 

on Friday
•

 

Best week since shutdown!
•

 

Dehumidifier unit “B”

 

is down, 2 
of 6 heater units had shut down
•

 

Due to the bonus set of 
needlenose

 

pliers Munters

 had included free of charge 
inside the heater coils

•

 

Running on “A”

 

alone while 
parts arrive for dehumidifier 
service
•Still better humidity than 
the pre-dehumidification era

- Sam Childress
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•

 

Horn 2 Leak Update:
•

 

Previously, decreased water 
flow for leaky Horn 2 line to 
40% of nominal (10 -> 4 gpm) 
•

 

Careful engineering 
evaluation

•

 

minimal effect on neutrino 
spectrum confirmed

•

 

Leak rate reduced by ~ change 
in flow, reduced rate of change 
of leak
•

 

Should gain several more 
weeks before intervention is 
necessary. 

•

 

Can not run with flow turned 
off.

Horn 2 Leak Rate (gal/day) 

- Sam Childress

Points to repairs starting end of March 
(after this furlough cycle) to optimize 
manpower resources and have spare horn 
ready if needed.  Est. 2+ weeks for repair.
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•
 

Consistent Protons/pulse:

~25x1012: 2+9 7-turns mixed

~30x1012: NuMI only (during 

pbar recycler transfers)
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Tue+Wed MI/NuMI accesses

TeV
F-sector access
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end of 2007 shutdown

Pr
ot
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)

Weekly Avg. needed to meet 2E20 pot by 1 September
Same, but with 4-weeks float (3 already spoken for!)
7-day running average



All Exp Mtg
Feb 11, 2008

NuMI-MINOS Status Report – Alec Habig

end of 2007 
shutdown
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